Author's response to reviews

Title: Atypical presentation of hepatocellular carcinoma: a mass on the left thoracic wall

Authors:

Sahin Coban (scoban72@yahoo.com)
Osman Yuksel (ottomanyuksel@hotmail.com)
Seyfettin Koklu (gskoklu@yahoo.com)
Koray Ceyhan (korayceyhan@yahoo.com)
Meltem Baykara (meltembaykara@yahoo.com)
Abdulkadir Dokmeci (akdokmeci@hotmail.com)

Version: 4 Date: 3 November 2004

Author's response to reviews:

Iratxe Puebla,
Assistant Editor
BioMed Central Middlesex House 34-42
Cleveland Street
London W1T 4LB
3 November, 2004
Dear Puebla;

1- We have revised our manuscript (MS: 2015234234206498 'Atypical Presentation of Hepatocellular Carcinoma: A Mass on the Left Thoracic Wall') in light of the reviewers' comments and made the changes according to your journal's format.

2- The manuscript was edited by someone who has native English.

3- As we mentioned in the manuscript, the patient died. Therefore, a written consent was obtained from the patient's relatives for publication of the patient's details. We have sent it by fax.

Please do not hesitate if you have any further suggestions.

Sincerely
Sahin Coban
Cukurambar Mah.
06520, Balgat,
Ankara, Turkey
Tel: +90 312 286 68 03
Fax: +90 312 310 63 71
E-mail: scoban72@yahoo.com